Author’s response to reviews

Title: The Rocks and Hard Places of MAiD: A qualitative study of nursing practice in the context of legislated assisted death

Authors:

Barbara Pesut (barb.pesut@ubc.ca)
Sally Thorne (sally.thorne@ubc.ca)
Catharine Schiller (catharine.schiller@unbc.ca)
Madeleine Greig (madeleine.greig@ubc.ca)
Josette Roussel (jroussel@cna-aiic.ca)

Version: 2 Date: 24 Jan 2020

Author’s response to reviews:

Dear Ms. Barling,

Thank you for the opportunity to revise this paper. We have included a section on abbreviations, deleted appropriate files, attached the interview guide, and made other revisions as requested. The uploaded manuscript is clean of all track changes and highlighting.

Warm regards, Barb Pesut